Author’s response to reviews

Title: A method for addressing research gaps in HTA, developed whilst evaluating robotic-assisted surgery: A proposal

Authors:

Luciana Ballini (luballini@regione.emilia-romagna.it)
Silvia Minozzi (minozzi.silvia@gmail.com)
Antonella Negro (anegro@regione.emilia-romagna.it)
Giampiero Pirini (gpirini@regione.emilia-romagna.it)
Roberto Grilli (rgrilli@regione.emilia-romagna.it)

Version: 4 Date: 6 August 2010

Author’s response to reviews:

the revised version sent few minutes ago did not contain the revised title and abstract, which have now been amended